**Author’s response to reviews**

**Title:** Knowledge brokering: Exploring the process of transferring knowledge into action

**Authors:**

Vicky L Ward (v.l.ward@leeds.ac.uk)  
Allan O House (a.o.house@leeds.ac.uk)  
Susan Hamer (s.hamer@leeds.ac.uk)

**Version:** 3  **Date:** 14 January 2009

**Author’s response to reviews:**

The title page has been restructured as requested:

Title
Author list
Affiliations
Email addresses
Correspondence address

The list of authors in the manuscript is written exactly as they are in the submission system.

Authors' qualifications (e.g. PhD / MD) have been removed from the manuscript and submissions system

Affiliations have been written in full and linked with the use of superscript numbers.

Email addresses have been included for all the authors on the title page.

The title has been removed from the head of the abstract.